Author's response to reviews

Title: A Case-Control Study of Determinants for High and Low Dental Caries Prevalence in Nevada Youth

Authors:

Marcia Ditmyer (marcia.ditmyer@unlv.edu)
Georgia Dounis (georgia.dounis@unlv.edu)
Connie Mobley (connie.mobley@unlv.edu)
Eli Schwarz (es@globaloralhealth.net)

Version: 3 Date: 31 October 2010

Author's response to reviews: see over
October 31, 2010

Tim Shipley, PhD
Executive Editor
BMC-series Journals
BioMed Central
Floor 6, 236 Gray's Inn Road
London, WC1X 8HL
email: editorial@biomedcentral.com

Dear Dr. Shipley,

Re: MS: 3571010640263895 - A Case-Control Study of Determinants for High and Low Dental Caries Prevalence in Nevada Youth

Attached are the details of the most recent revisions that were made to the manuscript.

1. Removed the tracking changes per request.

2. Added the references and tables into the original document per request. References following the manuscript and tables following the references.

3. Review and made 2 minor changes to the text. In the cover page, Dr. Dounis should be DDS rather than DDM and the title to Table 4 has been changed for clarity to:

   Table 4. Multiple Logistic Regression Results Demonstrating Odds-ratios of Being in the High DMFT Group vs. in the Cariesfree Group. All significant Independent Variables Remain in Final Model.

Thank you for all your help and support in this process. We are very excited about our publication. Should you have any questions, feel free to contact me at marcia.ditmyer@unlv.edu or in care of:

Marcia M Ditmyer, PhD, CHES
Assistant Professor, Department of BioMedical Sciences
UNLV School of Dental Medicine
1001 Shadow Lane Campus, MS 7425
Las Vegas, NV 89106

Regards,

Marcia M. Ditmyer, PhD, CHES
Marcia M. Ditmyer, PhD, CHES